Author’s response to reviews

Title: Sedentary lifestyle and Framingham risk scores: A population-based study in Riyadh city, Saudi Arabia

Authors:
AlJohara AlQuaiz (jalquaiz@yahoo.com)
Amna Siddiqui (amnarehana@gmail.com)
Ambreen Kazi (akamran@ksu.edu.sa)
Muhammad Batais (dmohammed34@gmail.com)
Ali AlHazmi (aalhazmii@ksu.edu.sa)

Version: 3 Date: 28 Feb 2019

Author’s response to reviews:

Dear Editor,

Thank you for your comments. We have made the required changes. The point mentioning "Please remove the " from the Competing Interests section, at the end of the sentence" is not clear, however we have shifted the sentence related to competing Interest to the end of the "Declaration" section. As suggested we have attached the questionnaire part related to this Manuscript. This survey "Women in Saudi Arabia health examination survey" was a large cross-sectional study addressing many objectives. Manuscripts from this survey related to Menopause, Violence, and vitamin D have been published elsewhere. The part related to the Framingham risk scores and the association with sedentary life style along with the questionnaire is being submitted to the journal BMC cardiovascular disorder.

We hope to receive a positive reply.

Thank you

Dr Ambreen Kazi

Corresponding Author